

To
The Editor in Chief
World Journal of Virology

07-11-2023

Dear Editor,

Enclosed, please find our revised manuscript (Manuscript NO: 89037) entitled “**Emerging Perspectives on RNA Virus-Mediated Infections: From Pathogenesis to Therapeutic Interventions**”. As suggested by the reviewers, we have revised the paper. The changes were highlighted. Our replies to the reviewers are as follow:

Need to address implications for microRNA pattern changes and roles in gliomas. If the above are addressed and references included (PMID: 35662986) , paper could be of interest.

Answer: We thank the reviewer for critical review and insightful comments. As per the suggestion we have modified manuscript and address implications for microRNA pattern changes and roles in gliomas with new headline.

Thank you very much

Best

Henu